Author’s response to reviews

Title: The development of a brief screener for autism using item response theory

Authors:

Caroline Mårland (caroline.marland@neuro.se)
Gitta Lubke (glubke@nd.edu)
Alessio Degl’ Innocenti (alessio.deglinnocenti@vregion.se)
Maria Råstam (maria.rastam@med.lu.se)
Christopher Gillberg (christopher.gillberg@gnc.gu.se)
Thomas Nilsson (thomas.nilsson@neuro.gu.se)
Sebastian Lundström (sebastian.lundstrom@neuro.gu.se)

Version: 1 Date: 13 Mar 2019

Author's response to reviews:

Please see the attached cover letter in which we have carefully considered the opinions of the two reviewers in a point-by-point manner.